Author's response to reviews

Title: Benefits of Whole Body Vibration Training in Patients Hospitalised for COPD Exacerbations - a Randomized Clinical Trial

Authors:

Timm Greulich (greulich@med.uni-marburg.de)
Christoph Nell (nell@med.uni-marburg.de)
Janine Koepke (koepke@staff.uni-marburg.de)
Juliane Fechtel (J.Fechtel@web.de)
Maja Franke (Maja.Franke@web.de)
Bernd Schmeck (schmeck@uni-marburg.de)
Daniel Haid (D.Haid1@gmx.de)
Sandra Apelt (Sandra.Apelt@uni-marburg.de)
Silke Filipovic (Silke.Mueller@med.uni-marburg.de)
Klaus Kenn (kkenn@schoen-kliniken.de)
Sabina Janciauskiene (sabinajanciauskiene@gmail.com)
Claus Franz Vogelmeier (claus.vogelmeier@med.uni-marburg.de)
Andreas Rembert Koczulla (koczulla@med.uni-marburg.de)

Version: 3
Date: 18 March 2014

Author's response to reviews:

Dear Editor!

Adressing the formal requests that we have been asked to correct in the above mentioned manuscript:

The clinical trial registration number was added to the manuscript. Furthermore, the email-adresses were added as requested.

Furthermore, we provided a point-to-point reply based on the reviewer's comments of Respiratory Research.

Thank you for considering the manuscript for publication in BMC Pulmonary Medicine.

A. R. Koczulla (MD)